Author’s response to reviews

Title: A systematic review protocol of timing, efficacy and cost effectiveness of upper limb therapy for motor recovery post-stroke.

Authors:

Kathryn Hayward (kathryn.hayward@florey.edu.au)
Sharon Kramer (sharon.kramer@unimelb.edu.au)
Vincent Thijs (vincent.thijs@florey.edu.au)
Julie Ratcliffe (Julie.Ratcliffe@unisa.edu.au)
Nick Ward (n.ward@ucl.ac.uk)
Leonid Churilov (leonidc@unimelb.edu.au)
Dale Corbett (dcorbett@uottawa.ca)
Geoffrey Cloud (g.cloud@alfred.org.au)
Laura Jolliffe (l.jolliffe@latrobe.edu.au)
Tina Kaffenberger (tina.kaffenberger@florey.edu.au)
Amy Brodtmann (agbrod@unimelb.edu.au)
Julie Bernhardt (j.bernhardt@florey.edu.au)
Natasha Lannin (n.lannin@alfred.org.au)

Version: 3 Date: 30 Jun 2019

Author’s response to reviews:

Thank you for again reviewing our paper and providing feedback to improve it. Below is a response to your comments.

Page 3. Abstract. Line 66. Where: "The outcomes will increase (…)" should read "We anticipate the findings of this review will increase (…)"

Response: The requested change has been made.

Response: The requested additional information has been added.